Author's response to reviews

Title: What is the 'problem' that outreach work seeks to address and how might it be tackled? Seeking theory in a primary health prevention programme.

Authors:

Mhairi Mackenzie (mhairi.mackenzie@glasgow.ac.uk)
Fiona Turner (fiona.turner@glasgow.ac.uk)
Stephen Platt (steve.platt@ed.ac.uk)
Maggie Reid (mcrten20@rocketmail.com)
Ying-ying Wang (yingying@sphsu.mrc.ac.uk)
Julia Clark (jclark54@caledonian.ac.uk)
Sanjeev Sridharan (sridharans@smh.toronto.on.ca)
Catherine A O'Donnell (kate.o'donnell@glasgow.ac.uk)

Version: 4 Date: 20 December 2011

Author's response to reviews:

Dear Editor - I am delighted that you are able to publish the attached article - I have made the major revisions but not the minor revisions as I am unable to do these prior to the deadline of 27th December. Best wishes
Mhairi Mackenzie